Author's response to reviews

Title: Sexual health of ethnic minority MSM in Britain (MESH project): design and methods

Authors:

Jonathan Elford (j.elford@city.ac.uk)
Eamonn McKeown (eamonn.mckeown.1@city.ac.uk)
Rita Doerner (rita.doerner.1@city.ac.uk)
Simon Nelson (simon.nelson@bristol.gov.uk)
Nicola Low (low@ispm.unibe.ch)
Jane Anderson (jane.anderson@homerton.nhs.uk)

Version: 2 Date: 29 June 2010

Author's response to reviews:

29 June 2010

Natalie Pafitis
The Biomed Central Editorial Team
BMC Public Health

Dear Natalie

MS: 1713696712403851
Sexual Health of ethnic minority MSM in Britain (MESH project): design and methods
Jonathan Elford, Eamonn McKeown, Rita Doerner, Simon Nelson, Nicola Low and Jane Anderson

Thank you for your email of 10 June 2010.

I have revised the manuscript as requested.

On page 6 of the manuscript, I have provided the aims of the study, within the Background section

On page 25 of the manuscript, I have provided a power calculation.

Both these changes are highlighted in yellow in the text

I hope these changes are acceptable and that you can now proceed with acceptance of the manuscript

Please don’t hesitate to contact me if you have any further queries

Thank you for your interest in this manuscript
Yours sincerely

Professor Jonathan Elford